EO-1 Weekly Status Week of March 20 – March 26, 2008
Day of Year 080 - 086
Mission Day 2684 - 2690
Earth Observing-One (EO-1) General

There were 109 Data Collection Events (DCEs) scheduled this week.  
On March 25, a telecon was held with the International Federation Red Cross (IFRC) and other international organizations working in the area of worldwide flooding disaster alerts.  The following significant items were discussed:

· Ukrainian Space Research Institute (USRI) representatives stated that their flood model can be expanded to other regions of interest.  The team agreed that the next region of interest be picked such that the IFRC and regional authorities can provide ground truth data to validate flood model output results.
· Discussion occurred about including the Dartmouth Flood Observatory data products in our collaborative experiment.
· Further coordination concerning the June 6th flood workshop at USRI in Kiev culminated in registration information being provided and an invitation to present individual team members’ component level view of flood activities.
· Still seeking input from the Japanese Aerospace Exploration Agency (JAXA) concerning availability of satellite data for our flood collaboration.
INSTRUMENTS
All instruments operated nominally this week.
Status of new ALI flood lamp calibration sequence to extend the data record turn-on and turn-off time period so as to capture the lamp warm-up and cool-down spectra.
· Upon complete review by the FOT the new sequence will be up-linked for execution. 

· Sequence currently planned to execute during the nominal GN contact.
· SGS S-Band contact to be on March 27 (087) at 13:11z.

· First command of the sequence to be executed on March 27 (087) at 13:15z.

Instrument decontamination cycles

· ALI deicing performed from March 25 (085) at 10:35z to March 26 (086) at 01:25z.

· HSI out-gassing performed from March 25 (085) at 10:45z to March 26 (086) 01:35z.

Two Lunar calibrations were performed on March 22 with the nominal lunar calibration performed during the 02:45-03:17z umbra and the modified lunar calibration performed during the 10:59-11:27z umbra.  The modified lunar calibration sequence allows centering the moon in the ALI and Hyperion field of views during the same event rather than alternating instrument centering every other month.
· FOT has changed the pitch offsets in the flight dynamics planning scripts to improve targeting.

· Results will be analyzed by the Goddard science team. 

· FOT will continue to work with the science team to further enhance the calibration sequence. 

· FOT also noticed low battery voltage readings upon completion of the second lunar calibration sequence.

· FOT will extend the gap between the two calibrations to at least three orbits to avoid possible harm to the batteries.  

EO-1 Spacecraft Subsystems
Command and Data Handling (C&DH)

· Continued to experience problems playing back engineering data from its Solid-State-Recorders.
Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

Sensor Web & Virtual Observatory Demonstrations

A teleconference was held between the Advanced Information System Technology (AIST) Sensor Web collaborators on Thursday, March 20.  The following items of interest were discussed:
· Two EO-1 Sensor Web movies were produced for the OWS-5 and RS-2008 meetings that show the fire, volcano, and flood sensor web activities.
· An EO-1 image was taken of the active Kilauea volcano which has dramatically produced a new gas vent in the crater.
· We are deriving additional data products for both the ALI and Hyperion under this activity and we discussed how to combine bands to produce specific new products for visible images, sulfur detection, snow/water/ ice/land discriminators, and clouds for example.
Autonomous Sciencecraft Experiment (ASE)

· The ASE controlled EO-1 all week.
EO-1 MISSION OPERATIONS CENTER
Real-Time
· AST – Autonomous Start Tracker invalid packets.

· On March 25 at 00:28z, FOT received a page that indicated bad house keeping temperature values for the AST.

· FOT reported to the MOC to further investigate the anomaly.

· Spacecraft reported invalid packet generation from the AST for a brief duration.

· It appears that the AST had gone through an acquisition cycle.

· FOT will investigate this anomaly to see if a reboot had occurred to cause generation of spurious temperature values.  

· Spacecraft state of health is currently nominal.

Mission Planning
No change from last report.
Flight Dynamics
No change from last report.
Data Analysis and Trending
No change from last report.
System Administration
No change from last report.
Operational Enhancements Requests and Discrepancy Reports
No change from last report.
GROUND AND SPACE NETWORK

Station Downtime
No change from last report.
Operational Discrepancies
No major problems to report.
Support Enhancements  
Another teleconference was held between the FOT and all PF-2 supporting parties on March 25, 2008 at 18:00z.
· A new scheduling test will be scheduled for early next week to test products in the control center that includes the PF-2 antenna.
· FOT will attempt to ingest the products into the mission planning systems.
· Upon completion of a successful scheduling test the FOT will schedule an engineering test to validate connections to the station and lights out operations.
· Station will also deliver the geodetics of the antenna to the MOC.
· FOT will introduce the new antenna to the flight dynamics systems. 
Upcoming Events
No major events are scheduled for the next two weeks.
Imagery Status
Scenes and Engineering Cals planned for week of March 20 – March 26, 2008              109
Total scenes and engineering calibrations planned for entire mission – approximately 39,008
Total Scenes:  ALI scenes in the Level 0 archive            35,424 (as of March 26, 2008)

                        Hyperion scenes in the Level 0 archive    35,173
Publications and Presentations Status (as of 12-05-07)
404 publications 

284 external presentations

53 articles and press releases
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